Author's response to reviews

Title: Simplified form of tinnitus retraining therapy in adults: a retrospective study

Authors:

Hashir Aazh (hashir.aazh@nhs.net)
Brian CJ Moore (bcjm@cam.ac.uk)
Brian R Glasberg (bg12@cam.ac.uk)

Version: 6 Date: 20 August 2008

Author's response to reviews: see over
Fig 1 "Decline of THI score"
It is not clear whether it represents % decline of initial score or absolute value. If later was the case since mean initial score was 60 and there is a number of cases with improvement above 60 (even close to 100) this would indicate large number of cases with final score close to zero. While it is possible I would like to be sure that indeed this is the case.

The decline in THI score is equal to pre-TRT score minus post-TRT score for each individual. I have clarified this in figure 1.